Preliminary An nou ncement

2 nd HELLENICCONFERENCE ON

ARTIFICIAL INTELLIGENCE( SETN - 2002)

11 - 12 APRIL 2002

TH ESSALONIK I, G REECE

URL: http://www.csd.auth.gr/~setn02

E - mail : setn02@csd.auth.gr

Scope and Obje ctive s

In recent timesthe fruits of research in the area of Ar tif icial Intelligence have f ound their way into various real - life applications, increasing thus the inter est and activity in the field. T he advent and rapid growth of the

Inter net and the World Wide Web have intensified this effect, through t he incr easing demand for intelligent information systems. The contribution of Gr eek researchers to this process has been substantial, both in

Greece and abroad. As a result, it seems to be the appr opriate time f or an event that will bring these researchers together in a common forum to exchange ideas and strengthen their collaborations. For this reason, the Hellenic Artificial Intelligence Society (EE T N), in cooperation with the Department of

Inf ormatics of the Aristotle University of Thessaloniki, isorgan ising the 2 nd Hellenic Conference on

Artificial Intelligence (SET N - 2002) in T hessaloniki, Apr il 11 - 12, 2002 .

T he conference pr ogram will consist of:

· Plenary and poster sessions, where research and application papers will be presented.

· A student session, where graduate students will be abl e to present their work in progress and receive feedback.

· Demonstrations of commer cial products and research prototypes.

Topics

Authors are invited to submit papers in any area of Artificial Intelligence including but not limited to:

· Knowledge Representat ion

· Reasoning and Search

· Planning and Scheduling

· Speech and Language Processing

· Machine Learning and Knowledge Discovery

· Neural Networks

· Logic Progr amming

· Computer Vision and Robotics

· Human Computer Interaction

· Multi - agent systems

· Applications of Intellig ent Systems

Or ganisation de tails

Language: Greek will be the official language of the conference. However, papers have to be written in

E nglish, aiming at a gr eater dissemination.

P roceedings: Plans are made for the proceedings of the conference to be published by an international publisher.

Registration: At least one author of each paper has to perfor m an early registr ation in the conference.

üü